Author's response to reviews

Title: Mosquito larvicidal activities of Solanum villosum berry extract against the dengue vector Stegomyia aegypti

Authors:

Nandita N Chowdhury (cnandita_2006@yahoo.co.in)
Anupam A Ghosh (anupamghosh75@yahoo.co.in)
Goutam G Chandra (goutamchandra63@yahoo.co.in)

Version: 5 Date: 8 January 2008

Author's response to reviews: see over
Sending herewith all the corrections made in the revised manuscript according to the instruction provided by you.

1. **Comments:** Page 2, Please delete the title and author list from this page.  
**Revision:** It has been deleted in the revised version of the manuscript.

2. **Comments:** Abstract - Please shorten the abstract to a maximum of 350 words.  
**Revision:** It has been revised and made upto a length of 344 words.

3. **Comments:** Result- Please rename this section Results.  
**Revision:** It has been renamed in the revised version.

4. **Comments:** Please include a ‘Competing interests’ section between the Conclusions and Authors’ contributions. If there are none to declare, please write ‘The authors declare that they have no competing interests’.  
**Revision:** In the revised manuscript it has been declared that we have no competing interests.

5. **Comments:** Please include an ‘Authors' contributions' section after the Competing interests.  
**Revision:** It has been included in the revised version of the text.

6. **Comments:** **Acknowledgements** – Please acknowledge anyone who contributed towards the study by making substantial contributions but who does not meet the criteria for authorship. Please also include their source(s) of funding. Please also acknowledge anyone who contributed materials essential for the study. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.
Revision: It has been included in the revised text. They do not provide any financial support to the work. We also obtain permission from them in this regard.

7. Comments: Please upload the figures as separate figure files using the “upload” form on the submission system, and delete the figure from the manuscript file.

Revision: It has been done in the revised format and the figures are now deleted from the text.